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Accumulated protons for past week
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Status of the experiment

● Both LH2 and LD2 targets were filled last week
● Target rotation is now only on those two targets 

and the empty flask
● St. 1 has a short and will be pulled for repair 

starting tomorrow. 
● Continuing to take data with the old St. 1 in 

place and operating.
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Computing summary


